School-based interventions to prevent anxiety, depression and conduct disorder in children and young people: a systematic review and network meta-analysis

Deborah M Caldwell,^{1*} Sarah R Davies,²
Joanna C Thorn,¹ Jennifer C Palmer,¹ Paola Caro,²
Sarah E Hetrick,³ David Gunnell,^{1,4}
Sumayya Anwer,⁵ José A López-López,⁶
Clare French,¹ Judi Kidger,¹ Sarah Dawson,¹
Rachel Churchill,⁵ James Thomas,⁷ Rona Campbell¹
and Nicky J Welton^{1,4}

Declared competing interests of authors: Sarah R Davies is the deputy managing editor for the Cochrane Psychosocial, Developmental and Learning Problems Review Group. Sarah E Hetrick is the joint co-ordinating editor of the Cochrane Common Mental Disorders Group and leads the Children and Young People Satellite group. Her position is part-funded by CureKids, a philanthropic organisation in New Zealand, and by Auckland Medical Research Foundation. David Gunnell and Nicky J Welton are supported by the National Institute for Health Research Biomedical Research Centre at University Hospitals Bristol NHS Foundation Trust and the University of Bristol.

Published July 2021 DOI: 10.3310/phr09080

¹Population Health Sciences, Bristol Medical School, University of Bristol, Bristol, UK ²School for Policy Studies, University of Bristol, Bristol, UK

³Faculty of Medical and Health Sciences, University of Auckland, Auckland, New Zealand

⁴National Institute for Health Research Bristol Biomedical Research Centre, Bristol, UK

⁵Centre for Reviews and Dissemination, University of York, York, UK

⁶Department of Basic Psychology and Methodology, Faculty of Psychology, University of Murcia, Murcia, Spain

⁷Evidence for Policy and Practice Information and Co-ordinating Centre (EPPI-Centre), University College London, London, UK

^{*}Corresponding author d.m.caldwell@bristol.ac.uk

Plain English summary

Interventions for anxiety, depression and conduct disorder Public Health Research 2021; Vol. 9: No. 8

DOI: 10.3310/phr09080

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

Anxiety, depression and conduct disorder are the most commonly diagnosed mental disorders among children and young people in the UK. Research suggests that preventing mental disorders from developing before adulthood can provide the largest benefit to the individual, society and the economy. Prevention programmes in schools are at the forefront of recent prevention attempts; studies evaluating whether or not they work have shown a small, but positive, effect.

In this report, we combined these studies to determine which type of school-based prevention programme was the most effective and best value for money for preventing anxiety, depression or conduct disorder. The types of programmes we included were psychological, educational and physical. For example, a physical intervention may be exercise, meditation or relaxation based. An educational intervention may provide information to the young person about mental health disorders and where to seek help. Psychological interventions typically address behavioural (actions and activities), cognitive (thoughts, reasoning, understanding), emotional and social factors. The programmes could be universal or targeted. Universal means that children are included regardless of whether or not they are showing signs of problems. Targeted means that only those children at higher risk of developing a problem, or already showing very early signs of mental health problems, are included.

When combining the results of studies, it is important that the studies include similar participants and comparable programmes, and record the effects of the programmes in similar ways. Programme effects are measured as 'outcomes' from the study. The main outcomes of interest in our report were symptoms of anxiety, depression and conduct disorder as reported by the young people themselves (self-reported). We were primarily interested in the outcomes immediately after the programme had been completed.

We separated studies into primary school settings (ages 4–11 years), secondary school settings (ages 12–18 years) and tertiary settings, for example university (up to 19 years of age), and planned separate statistical analyses for each. The findings were mixed. We found some evidence that in primary school settings cognitive–behavioural programmes may be effective in preventing symptoms 'of anxiety' but not symptoms of depression. In secondary school settings, universally delivered interventions based on cognitive–behavioural therapy and mindfulness or relaxation may be effective at preventing anxiety and depression. There was also evidence that exercise programmes may be effective when delivered to young people at higher risk (targeted) in secondary schools. We were not able to run similar analyses for the university settings. The studies evaluating prevention of conduct disorder were not similar enough to be combined and they did not use self-reported symptoms as their outcome measure. Instead, teachers and parents were asked to report on the students' behaviours. We did not run statistical analyses, but the authors of the original studies concluded that there was some evidence that programmes were effective in primary school settings.

Very few studies assessed the cost of the anxiety, depression or conduct disorder programmes, or whether or not they were value for money. Studies that did evaluate 'economic evidence' concluded that school-based, preventative interventions are unlikely to be value for money.

Many of the studies we included were small or not rigorously designed. Previous research has suggested that such studies are likely to overestimate the effectiveness of the interventions they evaluate. Therefore, we need to be cautious in interpreting the results of our study. Nevertheless, there was some evidence that school-based interventions are effective in preventing symptoms of anxiety, depression and conduct disorder. This evidence was weak, and we recommend that further large well-designed studies be conducted to investigate this further. Critically, these studies must also evaluate value for money.

Public Health Research

ISSN 2050-4381 (Print)

ISSN 2050-439X (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full PHR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/phr. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Public Health Research journal

Reports are published in *Public Health Research* (PHR) if (1) they have resulted from work for the PHR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Public Health Research* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

PHR programme

The Public Health Research (PHR) programme, part of the National Institute for Health Research (NIHR), is the leading UK funder of public health research, evaluating public health interventions, providing new knowledge on the benefits, costs, acceptability and wider impacts of non-NHS interventions intended to improve the health of the public and reduce inequalities in health. The scope of the programme is multi-disciplinary and broad, covering a range of interventions that improve public health.

For more information about the PHR programme please visit the website: https://www.nihr.ac.uk/explore-nihr/funding-programmes/public-health-research.htm

This report

The research reported in this issue of the journal was funded by the PHR programme as project number 15/49/08. The contractual start date was in October 2016. The final report began editorial review in January 2020 and was accepted for publication in December 2020. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The PHR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the PHR programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the PHR programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2021. This work was produced by Caldwell et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Scientific Adviser (Evidence Use), Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk